Author's response to reviews

Title: Introduction to the "Scoliosis" Journal Brace Technology Thematic Series: increasing existing knowledge and promoting future developments

Authors:

Stefano Negrini (stefano.negrini@isico.it)
Theodoros B Grivas (grivastb@gmail.com)

Version: 2 Date: 26 January 2010

Author's response to reviews:

This is an updated submission so that the title in the submission system reflects the thematic series and not a journal section. The document itself remains the same.